Craig Chambers CSE 501 (Winter ’95) Assignments

Reading Assignment, Week #3

For: Read: 1/16 holiday 1/18 [Alpern et al. 88] 1/20 [Ferrante et al. 87]

By: Do:

1/23 Write a new Cecil benchmark program, on the order of Towers of Hanoi, 8-queens, etc.

Send your source code (or a pointer to the source file(s)) by email to me and the TA’s.

Bibliography:

[Alpern et al. 88] Bowen Alpern, Mark N. Wegman, and F. Kenneth Zadeck. Detecting Equality of Variables in Programs. In Conference Record of the Fifteenth Annual ACM Symposium on Principles of Programming Languages, pp. 1-11, San Diego, CA, January, 1988.

[Ferrante et al. 87] Jeanne Ferrante, Karl J. Ottenstein, and Joe D. Warren. The Program Dependence Graph and Its Use in Optimization. In ACM Transactions on Pr ogramming Languages and Systems 9(3), pp. 319-349, July, 1987.